Author’s response to reviews

Title: Evidence for a Heritable Predisposition to Chronic Fatigue Syndrome

Authors:

Frederick Albright (frederick.albright@pharm.utah.edu)
Kathleen Light (kathleen.c.light@hsc.utah.edu)
Alan Light (alan.light@hsc.utah.edu)
Lucinda Bateman (lbateman@pol.net)
Lisa A Cannon-Albright (lisa.albright@utah.edu)

Version: 2 Date: 20 January 2011

Author's response to reviews: see over
Date: 19 Jan 2011

BioMed Central Ltd
Floor 6
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Re: Manuscript Submission, “Heritable Predisposition to Chronic Fatigue Syndrome”

To the Editor of Concern for Manuscript Submission, BMC Neurology,

We submit the attached paper referenced above to the New England Journal of Medicine for consideration for publication.

Human subject data is analyzed, but in a non-identifiable fashion, and with approved waiver of consent; rights of all subjects were protected.

Please use the information below for correspondence:

Frederick Albright, PhD
Pharmacotherapy Outcomes Research Center
421 Wakara Way, Suite 208
Salt Lake City, UT 84108
Email: frederick.albright@pharm.utah.edu
Phone: 801.587.9349
Fax: 801.587.7923

Thank you for your consideration,

Frederick Albright, Ph.D.
Research Assistant Professor
Department of Pharmacotherapy, College of Pharmacy, University of Utah
Salt Lake City, UT 84108
USA